Author's response to reviews

Title: Potential for early warning of viral influenza activity in the community by monitoring clinical diagnoses of influenza in hospital emergency departments

Authors:

Wei Zheng (wzhen@doh.health.nsw.gov.au)
Robert Aitken (raitk@doh.health.nsw.gov.au)
David J Muscatello (dmsc@doh.health.nsw.gov.au)
Tim Churches (tchur@doh.health.nsw.gov.au)

Version: 4 Date: 12 September 2007

Author's response to reviews: see over
The Editor  
BMC Public Health

Dear Sir/Madam

Potential for early warning of viral influenza activity in the community by monitoring clinical diagnoses of influenza in hospital emergency departments

I am writing to provide a revised manuscript in response to the email advice from the BMC Editorial Production Team re. formatting changes for the above manuscript.

As requested, we have made the following changes to the manuscript:

1. Use a hyphen when citing more than 2 consecutive references.

2. Change “Materials and Methods” to “Methods”.

3. Ensure the figures are cropped to minimise white space around the image.

I hope these changes are satisfactory and look forward to your response.

Yours sincerely

Wei Zheng  
Senior Surveillance Officer  
Centre for Epidemiology and Research  
12 September 2007